Author’s response to reviews

Title: Barriers and enablers in the implementation of a program to reduce cesarean deliveries

Authors:

Clara Bermúdez-Tamayo (clara.bermudez.easp@juntadeandalucia.es)
Emilia Fernández Ruiz (emi.fernandez.ruiz@gmail.com)
Guadalupe Pastor Moreno (pastormorenog@gmail.com)
Gracia Maroto-Navarro (gracia.maroto@juntadeandalucia.es)
Leticia García-Mochon (leticia.garcia@juntadeandalucia.es)
Francisco José Pérez-Ramos (fjpramos@gmail.com)
Maria del Pilar Vélez (maria.velez@queensu.ca)
Africa Caño-Aguilar (africacano59@gmail.com)

Version: 3 Date: 08 Jun 2017

Author’s response to reviews:

Reviewer reports:

Reviewer #2: The revision has addressed many of the reviewer's concerns. However, there is still a minor comment.

Authors have said that they have changed the dimensions in Table 4, but there is still the word "Factors" left.

On the other hand, though I can understand that with the use of qualitative methodology the items may be different to the those firstly exposed, it does not imply that the dimensions considered at the beginning of the study be changed too.

Reply:

- We have changed table 4, to include the word "Factors".
- Concerning to the dimensions of the study, we have explored the factors related, according to the literature (that we called "aspects to discuss" see table 1). Finally we have collected the barriers and enablers of each dimension, according to professionals views.